Author's response to reviews

Title: Prevalence of female genital mutilation and its effect on women's health in Bale zone, Ethiopia: a cross-sectional study

Authors:

Daniel Bogale (bogaleodo@gmail.com)
Desalegn Markos (desalegnmarkos@gmail.com)
Muhammedawel Kaso (muhammedawel@gmail.com)

Version: 4 Date: 6 October 2014

Author's response to reviews:

Method section is revised as per reviewers comments and RATS guideline was used to revise the manuscript